\documentclass[reqno]{amsart}
\usepackage{hyperref}

\AtBeginDocument{{\noindent\small
\emph{Electronic Journal of Differential Equations},
Vol. 2018 (2018), No. 177, pp. 1--9.\newline
ISSN: 1072-6691. URL: http://ejde.math.txstate.edu or http://ejde.math.unt.edu}
\thanks{\copyright 2018 Texas State University.}
\vspace{8mm}}

\begin{document}
\begin{thebibliography}{00}

\bibitem{agar1} R. P. Agarwal, L. Berezansky, E. Braverman, A. Domoshnitsky;
\emph{Nonoscillation Theory of Functional Differential Equations with Applications},
Springer, 2012. \MR 2908263

\bibitem{agar2} R. P. Agarwal, M. Bohner, Wan-Tong Li;
\emph{Nonoscillation and Oscillation: Theory for Functional Differential Equations},
Marcel Dekker, Inc., 2004. \MR 2084730

\bibitem{diblik0} J. Dibl\'ik;
\emph{A note on explicit criteria for the existence of positive solutions
to a linear advanced equation}, Appl. Math. Lett. 35 (2014) 72--76.

\bibitem{diblik1}  J. Dibl\'ik;
\emph{Integral criteria for the existence of positive solutions of
first-order linear differential advanced-argument equations},
Appl. Math. Lett., 63 (2017) 40--45.

\bibitem{diblik3} J. Dibl\'ik, M. K\'udel\v{c}\'ikov\'a;
\emph{New explicit integral criteria for the existence of positive solutions
to the linear advanced equation $x'(t)=c(t)x(t+\tau)$},
Appl. Math. Lett., 38 (2014) 144--148. \MR 3258218

\bibitem{diblik4} J. Dibl\'ik, K. Janglajew, M. K\'udel\v{c}\'ikov\'a;
\emph{New explicit coefficient criterion for the existence of positive
solutons to the linear advanced equations},
Discrete and Continuous Dynamical Systems B, 19, No. 8 (2014), 2461--2467.

\bibitem{elsgolts} L. E. Elsgolts, S. B. Norkin;
\emph{Introduction to the theory and application of differential equtions with
deviating arguments}, Academic Press, New York-London 1973.

\bibitem{erbe} L. H. Erbe, Q. Kong, B. G. Zhang;
\emph{Oscillation Theory for Functional Differential Equations},
Monographs and textbooks in pure and applied math, 190,
Marcel Dekker, 1994.

\bibitem{gyori}   I. Gy\"ori, G. Ladas;
\emph{Oscillation Theory of Delay Differential Equations},
Clarendon Press, 1991. \MR 1168471

\bibitem{ladas}   G. Ladas, I. P. Stavroulakis;
\emph{Oscillations caused by several retarded and advanced arguments},
J. Differential Equations, 44 (1) (1982) 134--152. \MR 0651691

\bibitem{zhang} B. G. Zhang;
\emph{Oscillation of the solutions of the first-order advanced type differential
 equations}, Sci. Explor., 2 (3) (1982) 79--82. (in Chinese) \MR 0713776

\end{thebibliography}

\title[\hfilneg EJDE-2018/177\hfil 
Existence of non-oscillatory solutions]
{Sufficient conditions for the existence of non-oscillatory solutions
to first-order differential equations with multiple \\ advanced arguments}

\author[J. G. Dix \hfil EJDE-2018/177\hfilneg]
{Julio G. Dix}

\address{Juio G. Dix \newline
Department of Mathematics,
Texas State University,
601 University Drive,
San Marcos, TX 78666, USA}
\email{jd01@txstate.edu}

\thanks{Submitted October 5, 2018. Published October 31, 2018.}
\subjclass[2010]{34K06, 34K11}
\keywords{Advanced linear differential equation; positive solution; 
\hfill\break\indent oscillatory solution} 

\begin{abstract}
 This article concerns the existence of
 non-oscillatory solutions to the equation
 \[
 x'(t)=\sum_{k=1}^m a_k(t) x(h_k(t)),
 \]
 where $a_k\geq 0$ and $h_k(t)\geq t$.
 We generalize existing results and then give an answer to the
 open question stated in \cite{diblik1}. Moreover we present a new condition
 based on the integral of $(\sum a_k)^2$.
\end{abstract}

\maketitle
\numberwithin{equation}{section}
\newtheorem{theorem}{Theorem}[section]
\newtheorem{lemma}[theorem]{Lemma}
\allowdisplaybreaks


\section{Introduction}

In this article we show sufficient conditions for the existence of
non-oscillatory solutions to the equation
\begin{equation} \label{e1}
x'(t)=\sum_{k=1}^m a_k(t) x(h_k(t))\,,
\end{equation}
where $a_k$, and $h_k$ are in $C([t_0,\infty),\mathbb{R})$ with $a_k(t)\geq 0$,
and $h_k(t)\geq t$.

Advanced equations have applications in real world problems where the current rate
of change may depend on future events. Such phenomena happen in population
dynamics and in economics, see \cite{elsgolts}. For additional information on
advanced equations see the books \cite{agar1, agar2,erbe,gyori, ladas,zhang}.

By a solution we mean a nontrivial function
$x\in C^1([t_0,\infty),\mathbb{R})$ that satisfies \eqref{e1}.
A solution is called oscillatory if it has arbitrarily large zeros;
otherwise is non-oscillatory. Note that if $x$ is a positive solution of \eqref{e1}
for sufficient large $t$,
then $-x$ is also a solution; so among non-oscillatory solutions, we only consider
positive solutions.

The main two approaches for solving \eqref{e1} are: using a fixed point argument,
and solving an inequality that implies the solvability of \eqref{e1}.
In this article we use the second approach. It is well known that
the existence of positive  solutions to \eqref{e1}  is equivalent to
the existence of a solution to the inequality
\begin{equation} \label{e2}
\lambda(t)\geq \sum_{k=1}^m a_k(t)\exp\Big(\int_t^{h_k(t)} \lambda(s)\,ds\Big) \quad
\forall t\geq t_0\,.
\end{equation}
However there is no specific way for finding solutions to \eqref{e2}.
Our strategy is to define solutions $\lambda$ of special form, and then find
the corresponding inequalities for $\sum a_k$ that guarantee the existence
of solutions to \eqref{e2}.
These inequalities can include values $\sum a_k(t)$, or integrals of
$\sum a_k$. In some publications, the
inequalities are called  pointwise and integral conditions, respectively.
Here, we generalize some of the existing conditions and then give an answer to the
open question stated in \cite{diblik1} (see inequality \eqref{e28}).
Also we present a new condition
based on the integral of $(\sum a_k)^2$.

The following theorem shows sufficient conditions for the existence of solutions to
\eqref{e1}; see \cite[Thm. 5.1]{agar1}, and for $m=1$ with $h_1(t)=t+\tau$,
see \cite[Thm. 1]{diblik4}.
The converse of the theorem  is easy to prove by using $\lambda(t)=x'(t)/x(t)$.

\begin{theorem} \label{thm1}
Suppose that inequality \eqref{e2} has a non-negative solution that is integrable
on each interval $[t_0,b]$. Then \eqref{e1} has a positive solution for $t\geq t_0$.
\end{theorem}

Note that letting
\begin{equation} \label{e3}
h(t)=\max\{h_k(t): k=1,2,\dots, m\}\,,
\end{equation}
it follows that solutions to
\begin{equation} \label{e4}
\lambda(t)\geq \sum_{k=1}^m a_k(t)\exp\Big(\int_t^{h(t)} \lambda(s)\,ds\Big) \quad
\forall t\geq t_0\,.
\end{equation}
are also solutions to \eqref{e2}.
Furthermore, if there is a constant $\tau>0$ such that
\begin{equation} \label{e5}
h(t)-t \leq \tau, \quad \forall  t\geq t_0\,,
\end{equation}
then solutions to
\begin{equation} \label{e6}
\lambda(t)\geq \sum_{k=1}^m a_k(t)\exp\Big(\int_t^{t+\tau} \lambda(s)\,ds\Big) \quad
\forall t\geq t_0
\end{equation}
are also solutions to \eqref{e4}, and hence to \eqref{e2}.


\section{Conditions using point values of $\sum a_k$}

 Initially we look for a constant solution to \eqref{e6}.
This leads to an extension of the well-known condition
$1/(\tau e)\geq a_1(t)$, see \cite{zhang}.

\begin{theorem} \label{thm2}
Under assumption \eqref{e5}, the condition
\begin{equation} \label{e8}
\frac{1}{\tau e}\geq \sum_{k=1}^m a_k(t)\quad \forall t\geq t_0\geq 1
\end{equation}
is sufficient for the existence of solutions to \eqref{e1}.
\end{theorem}

\begin{proof}
Let $\lambda(t)=\lambda_0$ be a constant. In this case, assuming  \eqref{e5},
condition \eqref{e6} is implied by
\[
\lambda_0\geq \Big(\sum_{k=1}^m a_k(t)\Big) e^{\lambda_0\tau}
 \quad \forall t\geq t_0\,.
\]
Note that the mapping $\lambda_0\mapsto \lambda_0/e^{\lambda_0\tau}$ has its
maximum at $\lambda_0=1/\tau$. With this value we obtain \eqref{e8} which
implies the existence of a solution to \eqref{e6}, and hence the existence of
 a positive solution to \eqref{e1}.
\end{proof}

Next we perturb the constant solution $1/\tau$, with a positive function that
tends  to zero as $t\to 0$ and leads to a simple integral in \eqref{e6}.

\begin{theorem} \label{thm3}
Under assumption \eqref{e5}, the condition
\begin{equation} \label{e10}
\frac{1}{\tau e}\big(1+\frac{\tau}{2t}\big)/\big(1+\frac{\tau}{t}\big)^{1/2}
\geq \sum_{k=1}^m a_k(t)\quad \forall t\geq t_0
\end{equation}
is sufficient for the existence of solutions to \eqref{e1}.
\end{theorem}

\begin{proof}
We look for solutions, to \eqref{e6}, of the form
$\lambda(t)=\frac{1}{\tau}+\frac{\alpha}{t}$,
where $\alpha\geq 0$ and $t\geq 1$. In this case \eqref{e6} is implied by
\[
\frac{1}{\tau}+\frac{\alpha}{t}
\geq \Big(\sum_{k=1}^m a_k(t)\Big)\exp\Big(\int_t^{t+\tau}
 \big(\frac{1}{\tau}+\frac{\alpha}{s}\big)\,ds
= \Big(\sum_{k=1}^m a_k(t)\Big)e\big(1+\frac{\tau}{t}\big)^\alpha
\]
which is equivalent to
\begin{equation} \label{e10b}
\frac{1}{\tau e}\big(1+\frac{\alpha\tau}{t}\big)
/\big(1+\frac{\tau}{t}\big)^\alpha
\geq \sum_{k=1}^m a_k(t) \quad\forall t\geq 1\,.
\end{equation}
Note that for $t\geq 1$ and $0\leq \alpha \leq 1$, using two terms of the Taylor
series for the mapping $x\mapsto (1+x)^\alpha$ about $x=0$, we have
\[
\big(1+\frac{\alpha\tau}{t}\big)/\big(1+\frac{\tau}{t}\big)^\alpha
\geq \big(1+\frac{\alpha\tau}{t}\big)
/\big(1+\frac{\alpha\tau}{t}\big) =1\,,
\]
with equality when $\alpha=0$ or $\alpha=1$.
Therefore \eqref{e10b} is less restrictive, on $\sum a_k$, than
\eqref{e8}.
Setting $\alpha=1/2$, we have condition \eqref{e10} which implies the
existence of solutions to \eqref{e6}; thus the existence of solutions to \eqref{e1}.
\end{proof}

Another perturbation of the constant solution $1/\tau$ is
$\lambda(t)= \frac{1}{\tau}+\frac{\alpha}{t^2}$ with $\alpha\geq 0$
and $t\geq 1$.
In this case assuming \eqref{e5}, condition \eqref{e6} is implied by
\[
\frac{1}{\tau}+\frac{\alpha}{t^2}
\geq \Big(\sum_{k=1}^m a_k(t)\Big)\exp\Big(\int_t^{t+\tau}
 \big(\frac{1}{\tau}+\frac{\alpha}{s^2}\big)\,ds\Big)
= \Big(\sum_{k=1}^m a_k(t)\Big)e\exp\Big(\frac{\alpha\tau}{t(t+\tau)}\Big)\,.
\]
Then the condition
\begin{equation} \label{e11}
\frac{1}{\tau e}\big(1+\frac{\alpha\tau}{t^2}\big)
/\exp\big(\frac{\alpha\tau}{t(t+\tau)}\big)
\geq \sum_{k=1}^m a_k(t) \quad\forall t\geq 1\,.
\end{equation}
is sufficient for the existence of a positive solution to \eqref{e1}.
Since
\[
\exp\big(\frac{\alpha\tau}{t(t+\tau)}\big)
\geq 1+ \frac{\alpha\tau}{t(t+\tau)}
\geq 1+\frac{\alpha\tau}{t^2}\,,
\]
the coefficient of $\frac{1}{\tau e}$ in \eqref{e11} is less than 1. Therefore,
\eqref{e11} is more restrictive, on $\sum a_k$, than \eqref{e8}.
Then there is no advantage in using \eqref{e11} instead of \eqref{e8}.

Next we consider the perturbed function
$\lambda(t)= \frac{1}{\tau}+\frac{\alpha}{t\ln(t)}$ with $\alpha\geq 0$
and $t\geq e$.
Under assumption \eqref{e5}, condition \eqref{e6} is implied by
\begin{align*}
\frac{1}{\tau}+\frac{\alpha}{t\ln(t)}
&\geq \Big(\sum_{k=1}^m a_k(t)\Big)\exp\Big(\int_t^{t+\tau}
 \big(\frac{1}{\tau}+\frac{\alpha}{s\ln(s)}\big)\,ds\Big)\\
&= \Big(\sum_{k=1}^m a_k(t)\Big)e\Big(\frac{\ln(t+\tau)}{\ln(t)} \Big)^\alpha\,.
\end{align*}
Then the condition
\begin{equation} \label{e12}
\frac{1}{\tau e}\Big(1+\frac{\alpha\tau}{t\ln(t)}\Big)
\big/\Big(\frac{\ln(t+\tau)}{\ln(t)}\Big)^\alpha
\geq \sum_{k=1}^m a_k(t) \quad\forall t\geq e\,.
\end{equation}
is sufficient for the existence of a positive solution \eqref{e1}.
There is no optimal value for $\alpha$ in \eqref{e12}, and if its exists, it depends
on $\tau$.

To compare \eqref{e12} with \eqref{e10} we graph the upper bounds
in both inequalities.
The graph corresponding to \eqref{e12} is eventually above the graph corresponding
to \eqref{e10} for each $\alpha$. This indicates that \eqref{e12} is less restrictive,
 on $\sum a_k$, than \eqref{e10}.

Next we consider the perturbed function
$\lambda(t)= \frac{1}{\tau}+\frac{\alpha}{t\ln(t)\ln(\ln(t))}$ with $\alpha\geq 0$
and $t\geq e^e$.
This leads to the condition
\begin{equation} \label{e14}
\frac{1}{\tau e}\Big(1+\frac{\alpha\tau}{t\ln(t)\ln(\ln(t))}\Big)
\big/\Big(\frac{\ln\big(\ln(t+\tau)\big)}{\ln\big(\ln(t)\big)}\Big)^\alpha
\geq \sum_{k=1}^m a_k(t) \quad\forall t\geq e^e\,.
\end{equation}
which is sufficient for the existence of a positive solution \eqref{e1}.
Graphing their upper bounds, we conclude that
\eqref{e14} is less restrictive than \eqref{e10}, but more restrictive than
\eqref{e12}.
Apparently further iterations of the logarithm do not lead to less restrictive
conditions.

Combining two of the perturbations above, we have
$\lambda(t)= \frac{1}{\tau} +\frac{1}{2t} +\frac{1}{2t\ln(t)}$.
For this $\lambda$,  we obtain the condition
\begin{equation} \label{e15}
\frac{1}{\tau e}\Big(1+\frac{\tau}{2t\ln(t)}\Big)
\big/\Big(\frac{\ln(t+\tau)}{\ln(t)}\Big)^{1/2}
\geq \sum_{k=1}^m a_k(t)\,.
\end{equation}
Using the the same $\lambda$  and Taylor polynomials, Diblik \cite{diblik0}
obtained the condition
\begin{equation} \label{e15b}
\frac{1}{\tau e}\Big(1+\frac{\tau^2}{8t^2}\Big)
\geq a_1(t)\,.
\end{equation}
Comparing the graphs of their bounds, we realize that \eqref{e15} is less restrictive
than \eqref{e15b}.
Some other functions $\lambda$ have been considered in \cite{diblik4},
such as
\[
\lambda(t)= \frac{1}{\tau}\Big(1+\frac{\tau}{2t}
+\frac{\tau}{2t\ln(t)} +\dots
+\frac{\tau}{2t\ln(t)\ln(\ln(t))\cdots \ln(\dots\ln(t))}\Big)\,.
\]

We conclude this section by stating that there are many good choices for $\lambda$,
but there is no known optimal choice.


\section{Conditions using weighted integrals of $\sum a_k$}

Initially we look for solutions to \eqref{e4},
of the form $\lambda(t)=\sum_{k=1}^m a_k(t) e^\delta$, where $\delta$ is a constant.
Under assumption \eqref{e3}, condition \eqref{e4} is implied by
\[
e^\delta \Big(\sum_{k=1}^m a_k(t)\Big)
\geq \Big(\sum_{k=1}^m a_k(t)\Big)\exp\Big(e^\delta \int_t^{h(t)}
\sum_{k=1}^m a_k(s)\,ds\Big)
\]
which is equivalent to
\[
\frac{\delta}{e^\delta} \geq \int_t^{h(t)} \sum_{k=1}^m a_k(s)\,ds\,.
\]
Note that the mapping $\delta \mapsto \delta/e^\delta$ has its maximun when
$\delta=1$. Therefore we set
\begin{equation} \label{e20}
\frac{1}{e} \geq \int_t^{h(t)} \sum_{k=1}^m a_k(s)\,ds\quad \forall t\geq t_0
\end{equation}
as a sufficient condition for the existence of positive solutions for
\eqref{e1}. This condition is the same as in \cite[Corollary 5.1]{agar1}.
Also when $m=1$ and $h(t)=t+\tau$, condition \eqref{e20} becomes the classical
condition $1/e \geq \int_t^{t+\tau} a_1(s)\,ds$.
It is also well known that if
$\liminf_{t\to\infty} \int_t^{t+\tau} a_1(s)\,ds> 1/e$, then
every solution of \eqref{e1} with $m=1$ is oscillatory;
see \cite[p. 31]{erbe}

Now we consider a perturbation of the solution
 $\lambda(t)=\sum_{k=1}^m a_k(t) e$, by using a non-negative function $w$ that decays
to zero as $t\to\infty$.  Let
\[
\lambda(t)=\sum_{k=1}^m a_k(t) e^{1+w(t)}\,.
\]
Then under assumption \eqref{e3}, condition \eqref{e4} is implied by
\[
\Big(\sum_{k=1}^m a_k(t)\Big) e^{1+w(t)}
\geq \Big(\sum_{k=1}^m a_k(t)\Big)
 \exp\Big(\int_t^{h(t)} e^{1+w(s)}\sum_{k=1}^m a_k(s)\,ds\Big)
\]
which is equivalent to the condition
\begin{equation} \label{e22}
\frac{1}{e}\big(1+w(t)\big)
\geq \int_t^{h(t)} e^{w(s)} \sum_{k=1}^m a_k(s)\,ds\quad \forall t\geq t_0\,,
\end{equation}
which correspond to \cite[Lemma 1]{diblik1}.
This is a sufficient condition, but it is an implicit condition.
With the goal of  separating the variables $w$ and $\sum a_k(s)$ in the above
inequality, we assume that $w$ is non-decreasing. Then we obtain the condition
\begin{equation} \label{e24}
\frac{1}{e}\big(1+w(t)\big)/ e^{w(t)}
\geq \int_t^{h(t)} \sum_{k=1}^m a_k(s)\,ds\quad \forall t\geq t_0\,.
\end{equation}
The variables are separated, but \eqref{e24} is more restrictive than \eqref{e22},
 because of the inequality $1+w\leq e^w$.

Next we follow the approach in \cite{diblik1} of partitioning the interval
$[t,t+\tau]$ into $n$ subintervals of equal length, $\tau/n$.

\begin{theorem} \label{thm4}
Assume \eqref{e4}, $n\geq 2$, and $w$ is non-increasing.
Also assume that there exists a non-increasing function $\beta_{n,\tau}$ such that
\begin{gather}
\int_t^{t+\tau/n}\sum_{k=1}^m a_k(s)\,ds
\leq \beta_{n,\tau}(t), \label{e26}\\
\intertext{and}
\frac{1}{e}\big(1+w(t)\big)
\geq \beta_{n,\tau}(t+\frac{\tau}{n})
 \sum_{i=1}^n \exp\big(w(t+\frac{(i-1)\tau}{n})\big), \label{e26b}
\end{gather}
for all $t\geq t_0$.
Then \eqref{e22} is satisfied so that there exists a positive solution to \eqref{e1}.
\end{theorem}

\begin{proof}
Note that the right-hand side of \eqref{e22} satisfies
\begin{align*}
\int_t^{t+\tau}e^{w(s)}\sum_{k=1}^m a_k(s)\,ds
&=\sum_{i=1}^n  \int_{t+(i-1)\tau/n}^{t+i\tau/n}e^{w(s)}\sum_{k=1}^m a_k(s)\,ds \\
&\leq \sum_{i=1}^n  \exp\big(w(t+(i-1)\tau/n)\big)
\int_{t+(i-1)\tau/n}^{t+i\tau/n}\sum_{k=1}^m a_k(s)\,ds \\
&\leq  \beta_{n,\tau}(t+\frac{\tau}{n})
\sum_{i=1}^n  \exp\big(w(t+\frac{(i-1)\tau}{n})\big)\,.
\end{align*}
This inequality and  \eqref{e26b} imply \eqref{e22}; thus there exists
a positive solution to \eqref{e1}.
\end{proof}

A concrete example of Theorem \ref{thm4} was obtained in \cite{diblik1},
where
\[
\beta_{n,\tau}(t)=\frac{1}{ne} + \frac{\mu(n-1)^2\tau^2}{8n^3t^2e}\quad
\text{and}\quad w(t)=\frac{(n-1)\tau}{2nt}\,,
\]
with $0<\mu<1$ and $n\geq 2$.

Trying to address the case $n=1$, Diblik \cite{diblik1} formulated the open question:
Prove of disprove that if for some $\mu\in(0,1)$ and $t_0>0$,
\begin{equation} \label{e28}
\sup_{t\geq t_0} t^2\Big(\int_t^{t+\tau}  a_1(s)\,ds-\frac{1}{e}\Big)
\leq \frac{\mu \tau^2}{8e},
\end{equation}
then there exists a positive solution to \eqref{e1} with $m=1$ and
$h(t)=t+\tau$.
A positive answer to this question is found as a particular case
of the next Theorem. First we rewrite \eqref{e28} as \eqref{e30} below.

\begin{theorem} \label{thm5}
Assume that \eqref{e5} holds and
\begin{equation} \label{e30}
\int_t^{t+\tau} \sum_{k=1}^m a_k(s)\,ds
\leq \frac{1}{e}\Big(1+\frac{\mu\tau^2}{8t^2}\Big)
\end{equation}
for some $\mu \in (0,1)$ and all $t\geq t_0$.
Then there exist a positive solution to \eqref{e1}.
\end{theorem}

\begin{proof}
Our strategy is to show that \eqref{e10} is satisfied, so that Theorem \ref{thm3}
can be applied.
Since $\sum a_k$ is a continuous function, by the mean value theorem for integrals,
there exists $\xi \in [t,t+\tau]$
such that
\[
\int_t^{t+\tau} \sum_{k=1}^m a_k(s)\,ds
=\tau\sum_{k=1}^m a_k(\xi)\,.
\]
Note that $\xi$ depends on $t,\tau,a_k$.
Then by \eqref{e30}, and since $t\leq \xi\leq t+\tau$, we have
\[
\sum_{k=1}^m a_k(\xi)
\leq \frac{1}{e\tau}\Big(1+\frac{\mu\tau^2}{8\xi^2}\Big)
\leq \frac{1}{e\tau}\Big(1+\frac{\mu\tau^2}{8(t-\tau)^2}\Big)\,.
\]
To show that $\sum a_k$ satisfies \eqref{e10}, we need to show that
\begin{equation} \label{e32}
\frac{1}{\tau e}\big(1+\frac{\tau}{2t}\big)/\big(1+\frac{\tau}{t}\big)^{1/2}
\geq \frac{1}{e\tau}\Big(1+\frac{\mu\tau^2}{8(t-\tau)^2}\Big)\,.
\end{equation}
Squaring both sides, this inequality is equivalent to
\begin{align*}
&1+\frac{\tau}{t}+\frac{\tau^2}{4t^2}\\
&\geq 1+\frac{\tau}{t}+\frac{\mu\tau^2}{4(t-\tau)^2}
+\frac{\mu\tau^3}{4t(t-\tau)^2}
+\frac{\mu^2\tau^4}{64(t-\tau)^4}
+\frac{\mu^2\tau^5}{64t(t-\tau)^4}\,.
\end{align*}
Note that
\[
\frac{\tau^2}{4t^2}-\frac{\mu\tau^2}{4(t-\tau)^2}
=\frac{\tau^2}{4}\Big(\frac{1-\mu}{(t-\tau)^2}
-\frac{2\tau}{t(t-\tau)^2} + \frac{\tau^2}{t^2(t-\tau)^2}\Big)\,.
\]
Since $\frac{1-\mu}{(t-\tau)^2}>0 $ and its decay is slower that the other two terms
as $t\to\infty$, the above inequality is true for $t$ large enough.
Therefore $\sum a_k$ satisfies \eqref{e10}, and by Theorem \ref{thm3}
there exists a positive solution to \eqref{e1}.
The proof is complete.
\end{proof}

We conclude this section by remarking that \eqref{e30} is less restrictive than
\eqref{e20}. Also for some examples the pair \eqref{e26}-\eqref{e26b}
is less restrictive than \eqref{e30}, but for other examples is more restrictive.

\section{Conditions using integrals of $(\sum a_k)^2$}

In this section we define solutions of \eqref{e4} depending on $\sum a_k$
and obtain integral conditions for $(\sum a_k)^2$.

\begin{theorem} \label{thm6}
Under assumption \eqref{e3}, the existence of a non-negative and continuous
function $g$  such that
\begin{equation} \label{e42}
g^2(t)/\Big(\int_t^{h(t)} e^{2g(s)}\,ds\Big)
\geq \int_t^{t+\tau} \Big(\sum_{k=1}^m a_k(s)\Big)^2 \,ds \quad \forall t\geq t_0
\end{equation}
implies the existence of a positive solution to \eqref{e1}.
\end{theorem}

\begin{proof}
Letting
$\lambda(t)=\sum_{k=1}^m a_k(t)e^{g(t)}$,
condition \eqref{e4} is implied by
\[
\Big(\sum_{k=1}^m a_k(t)\Big) e^{g(t)}
\geq \Big(\sum_{k=1}^m a_k(t)\Big)
\exp\Big(\int_t^{h(t)} \sum_{k=1}^m a_k(s) e^{g(s)} \,ds\Big)
\]
which is equivalent to
\[
g(t)\geq  \int_t^{h(t)} e^{g(s)}\sum_{k=1}^m a_k(s)  \,ds\,.
\]
From \eqref{e42} and the Cauchy-Schwarz inequality, we have
\[
g^2(t)
\geq  \Big(\int_t^{h(t)} e^{2g(s)}\,ds\Big)
\Big(\int_t^{h(t)} \big(\sum_{k=1}^m a_k(s)\big)^2  \,ds\Big)
\geq \Big(\int_t^{h(t)} e^{g(s)} \sum_{k=1}^m a_k(s) \,ds\Big)^2\,.
\]
Therefore $\lambda(t)$ is a solution of \eqref{e4}, which implies the existence of a
positive solution to \eqref{e1}.
\end{proof}

Now we consider $g(t)=R$, where $R>0$. Then \eqref{e42} is implied by
\[
\frac{R^2}{(h(t)-t)e^{2R}} \geq \int_t^{h(t)} \big(\sum_{k=1}^m a_k(s)\big)^2\,ds\,.
\]
Assuming \eqref{e5} and recalling that $R/e^R \leq 1/e$ for all $R$,
we have the condition
\begin{equation} \label{e44}
\frac{1}{\tau e^2} \geq \int_t^{t+\tau} \big(\sum_{k=1}^m a_k(s)\big)^2\,ds\quad
\forall t\geq t_0\,.
\end{equation}
This is a sufficient condition for the existence of a positive solution to
\eqref{e1}, because $\lambda(t)=\sum_{k=1}^m a_k(t)e$ is a solution of \eqref{e4}.

Now we consider the perturbed function $g(t)=1+\frac{1}{2} \ln(1+\tau/t)$.
Then
\[
\int_t^{t+\tau} e^{2g(s)}\,ds
= \int_t^{t+\tau} e^2(1+\tau/s)\,ds
 = \tau e^2\Big(1+\ln\big(1+\tau/t\big)\Big)\,.
\]
Following the above process we set
\begin{equation} \label{e45}
\frac{1}{\tau e^2}\big(1+\frac{1}{2} \ln(1+\tau/t)\big)^2/\big(1+\ln(1+\tau/t)\big)
\geq  \int_t^{t+\tau} \big(\sum_{k=1}^m a_k(s)\big)^2\,ds\quad
\forall t\geq t_0
\end{equation}
as a sufficient condition for the existence of a positive solution to \eqref{e1}.
Note that expanding the square above, we have
\begin{align*}
\big(1+\frac{1}{2} \ln(1+\tau/t)\big)^2
&=1+\ln(1+\tau/t)+\big(\frac{1}{2}\ln(1+\tau/t)\big)^2 \\
&\geq 1+\ln(1+\tau/t)\,.
\end{align*}
Therefore the coefficient of $1/(\tau e^2)$ in \eqref{e45} is greater than $1$;
thus \eqref{e45} is less restrictive than \eqref{e44}.

Now we consider the perturbed function
$g(t)=1+\frac{1}{2} \ln\big(1+\tau/(t\ln(t)\big)$.
Then
\[
\int_t^{t+\tau} e^{2g(s)}\,ds
= \int_t^{t+\tau} e^2(1+\tau/(s\ln(s))\,ds
= \tau e^2\Big(1+\ln\big(\frac{\ln(t+\tau)}{\ln(t)}\big)\Big)\,.
\]
Following the above process we set
\begin{equation} \label{e45b}
\frac{1}{\tau e^2}\Big(1+\frac{1}{2} \ln\big(1+\tau/(t\ln(t)\big)\Big)^2
/\Big(1+\ln\big(\frac{\ln(1+\tau)}{\ln(t)}\big)\Big)
\geq  \int_t^{t+\tau} \big(\sum_{k=1}^m a_k(s)\big)^2\,ds
\end{equation}
as a sufficient condition for the existence of a positive solution to \eqref{e1}.
Graphing their  upper bounds we notice that \eqref{e45} is less restrictive
than \eqref{e45b}.
As an example of a function that satisfies \eqref{e45b} but not \eqref{e45},
we have $a_1(t)=(1+\tau/(t^2)$,  when $m=1$ in \eqref{e1}.

We conclude this section with a theorem that uses $\int (\sum a_k)^2$ to
establish the oscillation of all solutions.

\begin{theorem} \label {thm7}
Assume that $m=1$, $0\leq a_1(t)\leq a_{\rm max}<\infty$ and $h(t)=t+\tau$ for all
$t\geq t_0$. If
\begin{equation} \label{e46}
\frac{1}{\tau e^2} < \liminf_{t\to \infty} \int_t^{t+\tau} a_1^2(s)\,ds\,,
\end{equation}
then all solutions to \eqref{e1} are oscillatory.
\end{theorem}

\begin{proof}
The strategy is to show that $1/e< \liminf_{t\to \infty} \int_t^{t+\tau} a_1(s)\,ds$
which implies the oscillation of all solutions.
From \eqref{e46}, there exist $t_1$ and $\epsilon>0$ such that
\[
\frac{1}{\tau e^2}+\epsilon\leq
\int_t^{t+\tau} a_1^2(s)\,ds \quad \forall t\geq t_1\,.
\]
Then
\begin{align*}
\epsilon&\leq \int_t^{t+\tau} \big(a_1^2(s) -\frac{1}{\tau^2e^2}\big)\,ds
=  \int_t^{t+\tau} \big(a_1(s) +\frac{1}{\tau e}\big)
\big(a_1(s)-\frac{1}{\tau e}\big)\,ds \\
&\leq (a_{\rm max}+\frac{1}{\tau e})
\int_t^{t+\tau} \big(a_1(s)-\frac{1}{\tau e}\big)\,ds
\end{align*}
Therefore,
\[
\frac{\epsilon}{(a_{\rm max}+\frac{1}{\tau e})}
\leq \int_t^{t+\tau} a_1(s)\,ds - \int_t^{t+\tau} \frac{1}{\tau e}\,ds
\quad \forall t\geq t_1\,,
\]
which implies
\[
\frac{1}{e} + \frac{\epsilon}{(a_{\rm max}+\frac{1}{\tau e})}
\leq  \liminf_{t\to \infty} \int_t^{t+\tau} a_1(s)\,ds\,.
\]
Thus, all solutions of \eqref{e1} are oscillatory.
\end{proof}

\subsection*{Acknowledgments}
I want to thank the anonymous referee for his/her comments that
improved this article, and Professor Zhaosheng Feng for managing my
submission to the EJDE.




\end{document}

